@ce UITF UITF status update — Oct 18/20

e Oct 13 - cleaned HD gas for Run 2 target to remove air (O, affects T, polarization lifetime)

e QOct 15 - condensed unpolarized HD; calibrated NMR response in Production Dewar (PD);

- installed HARP on beamline before IBC
- found that one of the 4 coils in the Rogowski coil BPM was shorted; repaired

e Oct 17 - cold 2K transfer of target to IBC with Transfer Cryostat

Goals for next week:

e complete simulations to optimize aperture sizes; machine apertures
e modify beamline before IBC: install beam-limiting apertures, ...

e tune NMR in IBC with HD target

(target will be cold-transferred out before the start of Run 2 (~ Oct 22"d)
and then transferred back after Raster studies)

e waiting to test new raster drivers
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